Author's response to reviews

Title: Risk of bias: A simulation study of power to detect study-level moderator effects in meta-analysis

Authors:

Susanne Hempel (susanne_hempel@rand.org)
Jeremy N.V. Miles (jeremy_miles@rand.org)
Marika J. Booth (sutterp@rand.org)
Zhen Wang (wang.zhen@mayo.edu)
Sally C. Morton (scmorton@pitt.edu)
Paul G. Shekelle (paul_shekelle@rand.org)

Version: 4
Date: 12 November 2013

Author’s response to reviews:

Dear Dr. Rodgers,
We have added the source of funding to the Acknowledgement section of the manuscript.
Yours sincerely,
Susanne Hempel
(on behalf of all authors)